Author’s response to reviews

Title: Is consumer behaviour towards footwear predisposing for lower extremity injuries in runners and walkers? A prospective study

Authors:

Tine Willems (tine.willems@ugent.be)

Roel De Ridder (roel.deridder@ugent.be)

Philip Roosen (philip.roosen@ugent.be)

Version: 1 Date: 15 Apr 2019

Author’s response to reviews:

Thank you very much for giving us the opportunity to adapt the manuscript.

As suggested by the editor, in the introduction, a reference is added to some sentences to support the factors identified.

Sentences with ‘…’ have been adapted and the manuscript has been reviewed by someone who is fluent in English.

Next to that, we have improved the description of the design of the questionnaire that was used in the study.